Author’s response to reviews

Title: Youth Depression Alleviation with Anti-inflammatory Agents (YoDA-A): A Randomised Clinical Trial of Rosuvastatin and Aspirin

Authors:

Michael Berk (mikebe@barwonhealth.org.au)
Patrick McGorry (pat.mcgorry@orygen.org.au)
Christopher Davey (christopher.davey@orygen.org.au)
Mohammadreza Mohebbi (m.mohebbi@deakin.edu.au)
Olivia Dean (o.dean@deakin.edu.au)
Sue Cotton (sue.cotton@orygen.org.au)
Andrew Chanen (andrew.chanen@orygen.org.au)
Seetal Dodd (seetald@barwonhealth.org.au)
Aswin Ratheesh (Aswin.Ratheesh@orygen.org.au)
G. Amminger (paul.amminger@orygen.org.au)
Mark Phelan (mark.phelan@orygen.org.au)
Amber Weller (amber.weller@orygen.org.au)
Andrew Mackinnon (andrew.mackinnon@biostats.com.au)
Francesco Giorlando (frank@greenant.net)
Shelley Baird (Shelley.Baird@orygen.org.au)
Lisa Incerti (lisa.incerti@orygen.org.au)
Rachel Brodie (rachel.brodie@orygen.org.au)
Natalie Ferguson (natalie.ferguson@orygen.org.au)
Simon Rice (simon.rice@orygen.org.au)
Miriam Schäfer (miriam.schaefer@orygen.org.au)
Author’s response to reviews:

1. Sup. tables S1-S6 are in appendix document – but tables S1-3 are also at the end of main text in manuscript document. Please ensure there are no duplications of files.

We thank the Editor for this observation and have included Tables S1-S6 in a separate document “Supplementary Appendix - Berk et al (2019) YoDA-A.docx” (included in this submission). Only Table 1 and Table 2 are now included in the main text of the manuscript document (pages 18 and 20 respectively). We have ensured there is no duplication of files.

2. Please upload all figures as separate files to the main manuscript

Figure 1 and Figure 2 have been uploaded as separate files to the main manuscript (“Figure 1.pdf” and “Figure 2.tif”). Figure titles are provided in the main manuscript (page 28).

AUTHOR NOTE:

IMPACT SRC at Deakin University has been renamed The Institute for Mental and Physical Health and Clinical Translation. Affiliations have been updated in the main manuscript.